Author’s response to reviews

Title: A clinically integrated Curriculum in Evidence-based Medicine for just-in-time learning through on-the-job training: The EU-EBM project

Authors:

Sjors FPJ Coppus (s.f.coppus@amc.uva.nl)
Jose I Emparanza (jemparan@chdo.osakidetza.net)
Julie Hadley (julie.hadley@bwhct.nhs.uk)
Regina Kulier (r.kulier@bham.ac.uk)
Susanne Weinbrenner (weinbrenner@azq.de)
Theodoros N Arvanitis (t.arvanitis@bham.ac.uk)
Amanda Burls (a.j.burls@bham.ac.uk)
Juan B Cabello (jbcabello@redcaspe.org)
Tamas Decsi (tamas.decsi@aok.pte.hu)
Andrea R Horvath (horvath@clab.szote.u-szeged.hu)
Marcin Kaczor (projektyue@caspin.org.pl)
Gianni Zanrei (gianni.zanrei@unicatt.it)
Karina Pierer (piererk@uhbs.ch)
Katarzyna Stawiarz (kstawiarz@caspin.org.pl)
Regina Kunz (rkunz@uhbs.ch)
Ben WJ Mol (b.w.mol@amc.uva.nl)
Khalid S Khan (k.s.khan@bham.ac.uk)

Version: 4 Date: 21 November 2007

Author’s response to reviews: see over
To the editor of BMC Medical Education

21 November 2007

Dear editor,

Thank you for acceptance of our manuscript.

We have made the following requested changes:
- line numbers have been removed
- full stop has been removed from title
- the manuscript is a clean version without track changes
- the reference style for the websites has been adjusted
- a conclusion section has been added at the end of the discussion section
- we have rename “conflict of interest statement” into “competing interests”
- table 1 has been moved to after the references
- figure legends have been moved to after the references. The figure replacement text has been deleted
- We have adjusted the reference to Table 1 in the text.

Table 1 has not been changed. It is a plane text table, which could be inserted as a ‘textbox’.

We hope the paper is now suitable for publication.

Yours sincerely,

On behalf of all co-authors,

Sjors FPJ Coppus, corresponding author

Academic Medical Centre
Department of Clinical Epidemiology and Biostatistics
Room J1B-216-1
Meibergdreef 9, 1105 AZ Amsterdam, The Netherlands.
E-mail: s.f.coppus@amc.uva.nl